\documentclass[a4paper]{panl}

\usepackage{cite}
\usepackage{wrapfig}
\usepackage{longtable}
\usepackage{rotating}
\usepackage{lscape}
\usepackage{multirow}
\usepackage{tabularx}
\usepackage{latexsym,graphicx,epsfig,psfrag,fancybox,xspace}
\usepackage{amssymb,amsmath,amsxtra,amsfonts}

%\theoremstyle{thmstyletwo}%
\newtheorem{example}{Example}%
\newtheorem{remark}{Remark}%
%\theoremstyle{thmstylethree}%
\newtheorem{definition}{Definition}%

\raggedbottom
\newcommand{\nn}{\nonumber\\}

\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\ena}{\end{eqnarray}}
\newcommand{\be}{\begin{equation}}
\newcommand{\en}{\end{equation}}
\newcommand{\Tr}{\mbox{\rm{tr}}}
\newcommand{\Br}{\ensuremath{\mathcal{B}}\xspace}
\newcommand{\ord}{\mathcal{O}}
\newcommand{\ds}{\displaystyle}
\newcommand{\bl}{\bigl}
\newcommand{\br}{\bigr}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}   
\newcommand{\Bla}{\Big\langle}
\newcommand{\Bra}{\Big\rangle}

\newcommand{\sll}{/\kern-4pt l}
\newcommand{\slp}{p\kern-5pt/}
\newcommand{\sls}{s\kern-5pt/}
\newcommand{\slell}{/\kern-5pt\ell}

\newcommand{\re}{{\rm Re}}
\newcommand{\im}{{\rm Im}}

\originalTeX
%\russianTeX
\begin{document}
\begin{thebibliography}{1}
\def\selectlanguageifdefined#1{
\expandafter\ifx\csname date#1\endcsname\relax
\else\selectlanguage{#1}\fi}
\providecommand*{\href}[2]{{\small #2}}
\providecommand*{\url}[1]{{\small #1}}
\providecommand*{\BibUrl}[1]{\url{#1}}
\providecommand{\BibAnnote}[1]{}
\providecommand*{\BibEmph}[1]{\emph{#1}}
\ProvideTextCommandDefault{\cyrdash}{\hbox to.8em{--\hss--}}
\providecommand*{\BibDash}{\ifdim\lastskip>0pt\unskip\nobreak\hskip.2em\fi
\cyrdash\hskip.2em\ignorespaces}

%%%%%%%%%%%%%%%%%%%INtroduction
%\cite{Belle:2007vmd}
\bibitem{Belle:2007vmd}
K.~F.~Chen \textit{et al.} [Belle],
%``Search for $B \to h^{(*)} \nu \bar\nu$ decays at Belle,''
Phys. Rev. Lett. \textbf{99}, 221802 (2007)
%doi:10.1103/PhysRevLett.99.221802
[arXiv:0707.0138 [hep-ex]].
%101 citations counted in INSPIRE as of 08 Nov 2022

%\cite{BaBar:2008wiw}
\bibitem{BaBar:2008wiw}
B.~Aubert \textit{et al.} [BaBar],
%``Search for $B \to K^{*} \nu \bar{\nu}$ decays,''
Phys. Rev. D \textbf{78}, 072007 (2008)
%doi:10.1103/PhysRevD.78.072007
[arXiv:0808.1338 [hep-ex]].
%47 citations counted in INSPIRE as of 08 Nov 2022

%\cite{BaBar:2010oqg}
\bibitem{BaBar:2010oqg}
P.~del Amo Sanchez \textit{et al.} [BaBar],
%``Search for the Rare Decay $B \to K \nu \bar{\nu}$,''
Phys. Rev. D \textbf{82}, 112002 (2010)
%doi:10.1103/PhysRevD.82.112002
[arXiv:1009.1529 [hep-ex]].
%74 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Belle:2013tnz}
\bibitem{Belle:2013tnz}
O.~Lutz \textit{et al.} [Belle],
%``Search for $B \to h^{(*)} \nu \bar{\nu}$ with the full Belle $\Upsilon(4S)$ data sample,''
Phys. Rev. D \textbf{87}, no.11, 111103 (2013)
%doi:10.1103/PhysRevD.87.111103
[arXiv:1303.3719 [hep-ex]].
%144 citations counted in INSPIRE as of 15 Nov 2022

%\cite{BaBar:2013npw}
\bibitem{BaBar:2013npw}
J.~P.~Lees \textit{et al.} [BaBar],
%``Search for $B \to K^{(*)} \nu \overline \nu$ and invisible quarkonium decays,''
Phys. Rev. D \textbf{87}, no.11, 112005 (2013)
%doi:10.1103/PhysRevD.87.112005
[arXiv:1303.7465 [hep-ex]].
%222 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Belle:2017oht}
\bibitem{Belle:2017oht}
J.~Grygier \textit{et al.} [Belle],
%``Search for $\boldsymbol{B\to h\nu\bar{\nu}}$ decays with semileptonic tagging at Belle,''
Phys. Rev. D \textbf{96}, no.9, 091101 (2017)
%doi:10.1103/PhysRevD.96.091101
[arXiv:1702.03224 [hep-ex]].
%198 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Belle-II:2021rof}
\bibitem{Belle-II:2021rof}
F.~Abudin\'en \textit{et al.} [Belle-II],
%``Search for B+\textrightarrow{}K+\ensuremath{\nu}\ensuremath{\nu}\textasciimacron{} Decays Using an Inclusive Tagging Method at Belle II,''
Phys. Rev. Lett. \textbf{127}, no.18, 181802 (2021)\newline
%doi:10.1103/PhysRevLett.127.181802
[arXiv:2104.12624 [hep-ex]].
%35 citations counted in INSPIRE as of 03 Nov 2022

%-------------  Theory ---------------------------------------------
%\cite{Altmannshofer:2009ma}
\bibitem{Altmannshofer:2009ma}
W.~Altmannshofer, A.~J.~Buras, D.~M.~Straub and M.~Wick,
%``New strategies for New Physics search in
%$B \to K^{*} \nu \bar{\nu}$, $B \to K \nu \bar{\nu}$ and
% $B \to X_{s} \nu \bar{\nu}$ decays,''
JHEP \textbf{04}, 022 (2009)
%doi:10.1088/1126-6708/2009/04/022
[arXiv:0902.0160 [hep-ph]].
%173 citations counted in INSPIRE as of 06 Apr 2021

%\cite{Buras:2021nns}
\bibitem{Buras:2021nns}
A.~J.~Buras and E.~Venturini,
%``Searching for New Physics in Rare $K$ and $B$ Decays without $|V_{cb}|$
%and $|V_{ub}|$ Uncertainties,''
Acta Phys. Polon. B \textbf{53}, no.6, A1
%doi:10.5506/APhysPolB.53.6-A1
[arXiv:2109.11032 [hep-ph]].
%21 citations counted in INSPIRE as of 05 Nov 2022

%\cite{Buras:2022wpw}
\bibitem{Buras:2022wpw}
A.~J.~Buras and E.~Venturini,
%``The exclusive vision of rare K and B decays and of the quark mixing
%in the standard model,''
Eur. Phys. J. C \textbf{82}, no.7, 615 (2022)
%doi:10.1140/epjc/s10052-022-10583-8
[arXiv:2203.11960 [hep-ph]].
%9 citations counted in INSPIRE as of 03 Nov 2022

%\cite{Parrott:2022dnu}
\bibitem{Parrott:2022dnu}
W.~G.~Parrott, C.~Bouchard and C.~T.~H.~Davies,
%``Standard Model predictions for $B\to K\ell^+\ell^-$,
%$B\to K\ell_1^- \ell_2^+$ and $B\to K\nu\bar{\nu}$ using form factors
%from $N_f=2+1+1$ lattice QCD,''
[arXiv:2207.13371 [hep-ph]].
%5 citations counted in INSPIRE as of 05 Nov 2022

%\cite{Kamenik:2009kc}
\bibitem{Kamenik:2009kc}
J.~F.~Kamenik and C.~Smith,
%``Tree-level contributions to the rare decays
%$B^+\to \pi^+ \nu \bar\nu$, $B^+\to K^+ \nu \bar\nu$,
%and $B^+\to K^{*\,+} \nu \bar\nu$ in the Standard Model,''
Phys. Lett. B \textbf{680}, 471-475 (2009)
%doi:10.1016/j.physletb.2009.09.041
[arXiv:0908.1174 [hep-ph]].
%49 citations counted in INSPIRE as of 06 Apr 2021

%\cite{Colangelo:1996ay}
\bibitem{Colangelo:1996ay}
P.~Colangelo, F.~De Fazio, P.~Santorelli and E.~Scrimieri,
%``Rare $B \to K^{(*)}$ neutrino anti-neutrino decays at $B$ factories,''
Phys. Lett. B \textbf{395}, 339-344 (1997)
%doi:10.1016/S0370-2693(97)00130-5
[arXiv:hep-ph/9610297 [hep-ph]].
%85 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Melikhov:1997zu}
\bibitem{Melikhov:1997zu}
D.~Melikhov, N.~Nikitin and S.~Simula,
%``Rare decays B ---\ensuremath{>} (K, K*) (lepton+ lepton-, neutrino anti-neutrino) in the quark model,''
Phys. Lett. B \textbf{410}, 290-298 (1997)\newline
%doi:10.1016/S0370-2693(97)00928-3
[arXiv:hep-ph/9704268 [hep-ph]].
%72 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Faessler:2002ut}
\bibitem{Faessler:2002ut}
A.~Faessler, T.~Gutsche, M.~A.~Ivanov, J.~G.~K\"orner and V.~E.~Lyubovitskij,
\newline
%``The Exclusive rare decays $B \to$ K(K*) $\bar{\ell} \ell$ and
%$B_c \to$ D(D*) $\bar{\ell} \ell$ in a relativistic quark model,''
Eur. Phys. J. direct \textbf{4}, no.1, 18 (2002)
%doi:10.1007/s1010502c0018
[arXiv:hep-ph/0205287 [hep-ph]].
%125 citations counted in INSPIRE as of 07 Nov 2022

%\cite{Wang:2012ab}
\bibitem{Wang:2012ab}
W.~F.~Wang and Z.~J.~Xiao,
%``The semileptonic decays $B/B_s \to (\pi, K)(\ell^+\ell^-,\ell\nu,\nu\bar{\nu})$ in the perturbative QCD approach beyond the leading-order,''
Phys. Rev. D \textbf{86}, 114025 (2012)
%doi:10.1103/PhysRevD.86.114025
[arXiv:1207.0265 [hep-ph]].
%101 citations counted in INSPIRE as of 07 Nov 2022

%\cite{Bause:2021cna}
\bibitem{Bause:2021cna}
R.~Bause, H.~Gisbert, M.~Golz and G.~Hiller,
%``Interplay of dineutrino modes with semileptonic rare B-decays,''
JHEP \textbf{12}, 061 (2021)
%doi:10.1007/JHEP12(2021)061
[arXiv:2109.01675 [hep-ph]].
%24 citations counted in INSPIRE as of 23 Nov 2022

%\cite{Issadykov:2022iwp}
\bibitem{Issadykov:2022iwp}
A.~Issadykov and M.~A.~Ivanov,
%``B \textrightarrow{} K(\ensuremath{*})\ensuremath{\nu}\ensuremath{\nu}̄ in covariant confined quark model,''
Mod. Phys. Lett. A \textbf{38} (2023) no.01, 2350006
doi:10.1142/S0217732323500062
[arXiv:2211.10683 [hep-ph]].

%\cite{Belle-II:2023esi}
\bibitem{Belle-II:2023esi}
I.~Adachi \textit{et al.} [Belle-II],
%``Evidence for B+\textrightarrow{}K+\ensuremath{\nu}\ensuremath{\nu}\textasciimacron{} decays,''
Phys. Rev. D \textbf{109} (2024) no.11, 112006
doi:10.1103/PhysRevD.109.112006
[arXiv:2311.14647 [hep-ex]].

%\cite{Issadykov:2022imz}
\bibitem{Issadykov:2022imz}
A.~Issadykov,
%``$B_{s}^{0} \to \bar {K}{\text{*}}{{(892)}^{0}}{{\ell }^{ + }}{{\ell }^{ - }}$ Decay in Covariant Confined Quark Model,''
Phys. Part. Nucl. Lett. \textbf{19} (2022) no.5, 460-462
doi:10.1134/S154747712205020X

%\cite{Soni:2021fky}
\bibitem{Soni:2021fky}
N.~R.~Soni, A.~Issadykov, A.~N.~Gadaria, Z.~Tyulemissov, J.~J.~Patel and J.~N.~Pandya,
%``Form factors and branching fraction calculations for $B_s \rightarrow D_s^{(*)} \ell ^+ \nu _\ell$ in view of LHCb observation,''
Eur. Phys. J. Plus \textbf{138} (2023) no.2, 163
doi:10.1140/epjp/s13360-023-03779-8
[arXiv:2110.12740 [hep-ph]].

%\cite{Soni:2020bvu}
\bibitem{Soni:2020bvu}
N.~R.~Soni, A.~Issadykov, A.~N.~Gadaria, J.~J.~Patel and J.~N.~Pandya,
%``Rare $b \rightarrow d$ decays in covariant confined quark model,''
Eur. Phys. J. A \textbf{58} (2022) no.3, 39
doi:10.1140/epja/s10050-022-00685-y
[arXiv:2008.07202 [hep-ph]].

%\cite{Issadykov:2023pwl}
\bibitem{Issadykov:2023pwl}
A.~Issadykov and S.~K.~Sakhiyev,
%``Bc\ensuremath{*} meson parameters and radiative decay width within the covariant confined quark model,''
Int. J. Mod. Phys. A \textbf{38} (2023) no.04n05, 2350040
doi:10.1142/S0217751X23500409
[arXiv:2307.05020 [hep-ph]].

%\cite{Dubnicka:2020xoh}
\bibitem{Dubnicka:2020xoh}
S.~Dubni\v{c}ka, A.~Z.~Dubni\v{c}kov\'a, A.~Issadykov, M.~A.~Ivanov and A.~Liptaj,
%``$Y(4260)$ as a four-quark state,''
Phys. Rev. D \textbf{101} (2020) no.9, 094030
doi:10.1103/PhysRevD.101.094030
[arXiv:2003.04142 [hep-ph]].

%\cite{Issadykov:2018myx}
\bibitem{Issadykov:2018myx}
A.~Issadykov and M.~A.~Ivanov,
%``The decays $B_{c}\to J/\psi+\bar\ell\nu_\ell$ and $B_{c}\to J/\psi + \pi(K)$ in covariant confined quark model,''
Phys. Lett. B \textbf{783} (2018), 178-182
doi:10.1016/j.physletb.2018.06.056
[arXiv:1804.00472 [hep-ph]].
%----------------- Xt ---------------------------------------------

%\cite{Misiak:1999yg}
\bibitem{Misiak:1999yg}
M.~Misiak and J.~Urban,
%``QCD corrections to FCNC decays mediated by Z penguins and W boxes,''
Phys. Lett. B \textbf{451}, 161-169 (1999)
%doi:10.1016/S0370-2693(99)00150-1
[arXiv:hep-ph/9901278 [hep-ph]].
%297 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Buchalla:1998ba}
\bibitem{Buchalla:1998ba}
G.~Buchalla and A.~J.~Buras,
%``The rare decays $K\to \pi \nu\bar\nu$, $B \to X \nu\bar\nu$ and $B \to l^+ l^-$: An Update,''
Nucl. Phys. B \textbf{548}, 309-327 (1999)\newline
%doi:10.1016/S0550-3213(99)00149-2
[arXiv:hep-ph/9901288 [hep-ph]].
%461 citations counted in INSPIRE as of 15 Nov 2022

%\cite{Brod:2010hi}
\bibitem{Brod:2010hi}
J.~Brod, M.~Gorbahn and E.~Stamou,
%``Two-Loop Electroweak Corrections for the $K \to \pi \nu \bar{\nu}$ Decays,''
Phys. Rev. D \textbf{83}, 034030 (2011)\newline
%doi:10.1103/PhysRevD.83.034030
[arXiv:1009.0947 [hep-ph]].
%358 citations counted in INSPIRE as of 15 Nov 2022

%------------- helicities --------------------

%\cite{Ivanov:2015tru}
\bibitem{Ivanov:2015tru}
M.~A.~Ivanov, J.~G.~K\"orner and C.~T.~Tran,
%``Exclusive decays $B \to \ell^-\bar\nu$ and $B \to D^{(\ast)}\ell^-\bar\nu$ in the covariant quark model,''
Phys. Rev. D \textbf{92}, no.11, 114022 (2015)
%doi:10.1103/PhysRevD.92.114022
[arXiv:1508.02678 [hep-ph]].
%50 citations counted in INSPIRE as of 15 Nov 2022



\end{thebibliography}

% Journal sections (see http://pkp.jinr.ru/index.php/PEPAN_LETTERS/about/editorialPolicies#focusAndScope)
%\issuearea{Physics of Elementary Particles and Atomic Nuclei. Theory}

\title{ $B \to K^{(*)} \nu \bar\nu$ decays: data and theoretical predictions\\ }

%\title{Распады $B \to K^{(*)} \nu \bar\nu$: экспериментальные данные и теоретические предсказания}

\maketitle
\authors{Aidos Issadykov\vspace{1mm} $^{a,b}$ \vspace{2mm}\footnote{E-mail: issadykov.a@gmail.com}}
%\authors{Айдос Исадыков$^{a,b}$\footnote{E-mail: issadykov.a@gmail.com (русский вариант)},}

\from{$^{a}$\,The Institute of Nuclear Physics, \\
Ministry of Energy of the Republic of Kazakhstan, 050032 Almaty,  {\it KAZAKHSTAN}} 
%\from{$^{a}$\,Институт ядерной физики Министерства Энергетики Республики Казахстан, 050032 Алматы, Казахстан

\from{$^{b}$\,Joint Institute for Nuclear Research, 141980 Dubna, \textit{RUSSIA}}
%\from{$^{b}$\,Объединенный институт ядерных исследований, 141980 Дубна, Россия 


\begin{abstract}
In this paper we compare data from Belle II and BaBar with available results of theoretical approaches for $B \to K^{(*)} \nu \bar\nu$ decays.
The theoretical predictions for $B^+ \to K^{+} \nu \bar{\nu}$ are lower than experimental data. This may indicate on the possibility of New physics effects in this decay.
\end{abstract}
\vspace*{6pt}

%\begin{abstract}
%В этой статье мы сравниваем данные от Belle II и BaBar с имеющимися результатами теоретических подходов для распадов $B \to K^{(*)} \nu \bar\nu$. Теоретические предсказания для $B^+ \to K^{+} \nu \bar{\nu}$ на порядок меньше экспериментальных данных. Это может указывать на возможность эффектов Новой физики в этом распаде.
%\end{abstract}

\noindent
12.40.$-$y; 14.40.$-$n

\label{sec:intro}
\section*{INTRODUCTION}
Decays involving flavor-changing neutral currents are of great interest for studying and uncovering new physical phenomena. In the decay process $B \to K^{(*)} \mu^+ \mu^{-}$, discrepancies have been observed between theoretical predictions and experimental data. Over the past several years, research studies have been carried out to detect the decay of B mesons to K mesons accompanied by a pair of neutrinos~\cite{Belle:2007vmd,BaBar:2008wiw,BaBar:2010oqg,Belle:2013tnz,BaBar:2013npw,Belle:2017oht,Belle-II:2021rof}.
The rare decays of B mesons to final states containing a pair of neutrinos are theoretically considered one of the cleanest processes involving neutral currents that facilitate flavor changes in quarks. The electrical neutrality of neutrinos enables an accurate separation of hadronic and lepton currents, unlike other B meson decays. Various theoretical frameworks have provided predictions for the branching fraction of this decay~\cite{Altmannshofer:2009ma,Buras:2021nns,Buras:2022wpw,Parrott:2022dnu,Kamenik:2009kc,Colangelo:1996ay,Melikhov:1997zu,Faessler:2002ut,Wang:2012ab,Bause:2021cna}.
The theoretical predictions for the $B^+ \to K^{+} \nu \bar{\nu}$ decay process obtained within the CCQM(Covariant Confined Quark Model) framework~\cite{Issadykov:2022iwp} are consistent with other theoretical methods, and were substantially lower than the experimental upper bounds known at the time. In late 2023, the Belle II collaboration reported the measured branching fraction for $B^+ \to K^{+} \nu \bar{\nu}$~\cite{Belle-II:2023esi}, providing data for the inclusive, hadronic, and combined decay modes.
\bea
\Br_{inc}(B^+\to K^+\nu\bar\nu) &=& \langle 2.7 \pm 0.5(stat) \pm 0.5(syst)\rangle 10^5
\nn
\Br_{hadr}(B^+\to K^+\nu\bar\nu) &=& \langle1.1^{+0.9}_{-0.8}(stat) ^{+0.5}_{-0.4}(syst)\rangle 10^5
\nn
\Br_{comb}(B^+\to K^+\nu\bar\nu) &=& \langle2.3 \pm 0.5(stat) ^{+0.5}_{-0.4}(syst)\rangle 10^5
\ena
In this paper, we reviewed theoretical predictions within the Standard Model and experimental data.


\label{sec:model}
\section*{Covariant confined quark model}
We will use the covariant confined quark model~\cite{Issadykov:2022iwp,Issadykov:2022imz,Soni:2021fky,Soni:2020bvu,Issadykov:2023pwl,Dubnicka:2020xoh,Issadykov:2018myx} in our calculations. This analysis is based on nonlocal interpolation currents that possess the relevant quantum numbers of hadrons. The effective Hamiltonian for $b\to s\nu\bar\nu$~transitions in the Standard Model can be expressed in the following form
%
\be
{\mathcal H}_{\rm eff}^{\rm SM} =
\frac {G_F}{\sqrt2}  V_{tb} \, V^*_{ts}
\Big[\frac {\alpha_{em}}{2\pi} \frac {X_t}{\sin^2\Theta_W}\Big]
(\bar s\, O_\mu\, b) (\bar\nu O^\mu \nu) + h.c.
\en
Here $G_F$  is the Fermi constant,  $V_{tb}$ and  $V^\ast_{ts}$
are the matrix elements of the Cabibbo-Kobayashi-Maskawa (CKM) matrix,
$\alpha_{em}$ is the fine structure constant calculated at the
electroweak scale of $Z$-boson mass,  $\theta_W$ is the Weinberg angle,
$O_\mu=\gamma_\mu(1-\gamma_5)$ is the weak Dirac matrix with the left
chirality. The Wilson coefficient $X_t$ is known with a high accuracy,
including NLO QCD corrections~\cite{Misiak:1999yg,Buchalla:1998ba} and
two-loop electroweak contributions~\cite{Brod:2010hi}, resulting in
%
\be
X_t=1.469 \pm 0.017.
\en
The matrix elements $\langle K^{(\ast)} | \bar s O^\mu b) | B \rangle$ can be expressed as scalar functions, known as form factors, which depend on the square of the transferred momentum
%
\bea
\langle K(p_2)\,|\,\bar s\, O^{\,\mu}\, b\,| B(p_1)\rangle
&=&
F_+(q^2)\, P^{\,\mu} + F_-(q^2)\, q^{\,\mu},
\label{eq:PP'}\\[1.5ex]
\langle K^\ast(p_2,\epsilon_2)\,|\,\bar s\, O^{\,\mu}\,b\, |\,B(p_1)\rangle
&=&
\frac{\epsilon^{\,\dagger}_{2\,\alpha}}{m_1+m_2}\,
\Big( - g^{\mu\alpha}\,Pq\,A_0(q^2) + P^{\,\mu}\,P^{\,\alpha}\,A_+(q^2)
\nn
&+& q^{\,\mu}\,P^{\,\alpha}\,A_-(q^2)
+ i\,\varepsilon^{\mu\alpha P q}\,V(q^2)\Big).
\label{eq:PV}
\ena
Here $P =p_1+p_2$, \,$q=p_1-p_2$, and $\epsilon_2$ is the polarization vector
of the $K^\ast$ meson, so $\epsilon_2^\dagger\cdot p_2 = 0$ .
The abbreviation
$\varepsilon^{\mu\alpha P q} = \varepsilon^{\mu\alpha \nu\delta} P_\nu q_\delta$
is adopted, where the absolutely antisymmetric Levi-Civita tensor is
defined as $\varepsilon^{0123} = -\varepsilon_{0123} = -1$.
The particles are on the mass shell, i.e.
$p_1^2=m_1^2=m_B^2$ and $p_2^2=m_2^2=m_{K^{(\ast)}}^2$.

The calculation of form factors necessitates approaches beyond the perturbation theory, such as quark models, QCD sum rules, and lattice QCD. Considering the aforementioned definitions of form factors, the differential branching fraction can be expressed as follows:
%
\begin{align}
\frac{ d \Br (B\to K^{(\ast)}+\nu\bar\nu)}{dq^2} &=
3\tau_{B} \frac{ (G_F \lambda_t \alpha_{em})^2 }{3(2\pi)^5}
\Big(\frac{X_t}{\sin^2\theta_W}\Big)^2
\nn
&\times \frac{|{\bf p_2}|}{4m_1^2}\,
\Big( {\tilde H}_+^2 + {\tilde H}_-^2 + {\tilde H}_0^2 \Big).
\label{eq:BR}
\end{align}
The factor 3 arises from the summation over the three neutrino flavors: $\nu_e$, $\nu_\mu$ and $\nu_\tau$. In the Standard Model,
they are assumed to be massless and therefore do not interfere with each other.
Additionally, $\tau_{B}$ is the B-meson lifetime,
$\lambda_t\equiv| V_{tb}V^\ast_{ts} |$,
$|{\bf p_2}|=\lambda^{1/2}(m_1^2,m_2^2,q^2)/2m_1 $ is the  daughter meson momentum
in the rest frame of the initial $B$~meson.
Scaled helicity amplitudes ${\tilde H}$
are related to the helicity amplitudes defined in \cite{Ivanov:2015tru}
as ${\tilde H} = \sqrt{q^2} H$. Consequently, we have found that\\[1.2ex]
%
\noindent
{\boldmath $B\to K$ \,\,\bf transition:}
\be
{\tilde H}_\pm = 0,
\qquad
{\tilde H}_0 = 2\,m_1\,|{\bf p_2}|\,F_+.
\label{eq:hel_PP}
\en
%
\noindent
{\boldmath$B\to K^\ast$ \,\,\bf transition:}
\bea
{\tilde H}_\pm &=&
\frac{\sqrt{q^2}}{m_1+m_2}
\Big(-Pq\, A_0 \pm 2\,m_1\,|{\bf p_2}|\, V \Big),
\nn[1.2ex]
{\tilde H}_0 &=&
\frac{1}{m_1+m_2}\frac{1}{2\,m_2}
\Big(-Pq\,(Pq - q^2)\, A_0 + 4\,m_1^2\,|{\bf p_2}|^2\, A_+\Big).
\label{eq:hel_PV}
\ena

The scaling is done in order to avoid uncertainties at the point $q^2 = 0$ in
numerical calculations. We note that in these decays the physical range of
the variable $q^2$ is $0\le q^2 \le q^2_{\rm max}=(m_1-m_2)^2$. The more information about model and parameters can be found in~\cite{Issadykov:2022iwp}.

\section*{Numerical results and conclusions}
\label{sec:results}
Table~\ref{tab:BR_tot} presents the results of the total branching fraction calculations for the $B \to K^{(\ast)}\nu\bar\nu$ process, as computed within the CCQM framework. For comparison, the table also includes the findings from experimental measurements and theoretical predictions derived using various approaches.

The theoretical predictions for the branching ratio of $B \to K\nu\bar\nu$ differs from the experimental measurement by approximately $2.4$ standard deviations.
%
%%%%%%%%%%%%%%%%%%
\begin{table}[ht]
\caption{Summary of experimental and theoretical results for the branching
  fractions $\mathcal{B}(B\to K^{(\ast)}\nu\bar{\nu})$. Experimental upper
  limits are given at 90\% confidence level.}
\label{tab:BR_tot}  
  \begin{center}
    \def\arraystretch{1.2}
\begin{tabular}{c|c|c}
  \hline
Ref &{$10^5\Br(B^+\to K^+\nu\bar\nu)$} &
{$10^5\Br(B\to K^{\ast}\nu\bar\nu)$} \\
\hline
BaBar~\cite{BaBar:2013npw} &$<1.6$  & $<6.4$  \\

Belle-II~\cite{Belle-II:2023esi,Belle:2017oht} &$2.3 \pm 0.5(stat) ^{+0.5}_{-0.4}(syst)$  & $<6.1$ \\
\hline
Our model~\cite{Issadykov:2022iwp} & 0.496(0.074) &  0.957(0.143) \\
\cite{Altmannshofer:2009ma} & 0.45(0.07) &  0.68$^{+0.10}_{-0.11}$  \\
 \cite{Buras:2021nns}&0.445(0.062) & 0.970(0.092) \\
  \cite{Buras:2022wpw}&0.465(0.062) & 1.013(0.092)  \\
 \cite{Kamenik:2009kc}& 0.51(0.08) &  0.84(0.14) \\
\cite{Colangelo:1996ay} &0.24(0.06)   & 0.51(0.08) \\
 \cite{Melikhov:1997zu}  &0.52(0.11) &  1.3(0.5)   \\
 \cite{Bause:2021cna} & 0.423(0.056) & 0.893(0.107) \\[1.2ex]
 \hline
\end{tabular}
\end{center}
\end{table}

\label{sec:funding}
\section*{Funding}

This research has been funded by the Committee of Science of the Ministry of Science and Higher Education of the Republic of Kazakhstan (Grant No. BR24992891). 


\end{document}
